Author’s response to reviews

Title: β-blockers after myocardial infarction and 1-year clinical outcome – a retrospective study

Authors:

Tora Hagsund (torahagsund@hotmail.com)
Sven-Erik Olsson (Sven-Erik.Olsson@skane.se)
Gustav J Smith (j.gustav.smith@gmail.com)
Bjarne Madsen Hardig (bjarne.madsen_hardig@med.lu.se)
Henrik Wagner (henrik.wagner@med.lu.se)

Version: 3 Date: 16 Mar 2020

Author’s response to reviews:

Dear Editor in Chief

Please find enclosed the final version without track changes of the manuscript entitled “β-blockers after myocardial infarction and 1-year clinical outcome – a retrospective study” by Tora Hagsund M, Sven-Erik Olsson MD, J. Gustav Smith MD PhD, Bjarne Madsen Hardig, RN PhD, and Henrik Wagner MD PhD (BCAR-D-19-00354).

We have addressed the final Technical Comments:

1. The e-mail for the corresponding author differs between the manuscript file and the Editorial Manager system. Please ensure that the same e-mail is used for both.

   Has been changed

2. In the Ethics approval and consent to participate section please cite the relevant national Swedish and European guidelines.

   Text has been added addressing this
3. In the Ethics approval and consent to participate section please clarify if consent was informed and if it was written or verbal.

Text has been added addressing this

4. In Funding, please state whether or not the funding body played any roles in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Text has been added addressing this

5. Upon resubmission, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published and remove any files that should not form part of the final manuscript (e.g. covering letters, responses to reviewers, etc). If you wish to respond to these revision requests, please insert the information in to the designated input box only.

This has been done.

Best Regards

Bjarne Madsen Hardig RN, PhD
Department of Cardiology
Lund University
22242 Lund
Sweden
Phone: +46 768 58 19 40
Mail: bjarne.madsen_hardig@med.lu.se